Author's response to reviews

Title: Valsalva retinopathy in pregnancy

Authors:

Abdullah S Al-Mujaini (mujainisqu@hotmail.com)
Carolina C Montana (caro_montana@yahoo.com)

Version: 2 Date: 22 January 2008

Author's response to reviews: see over
Jan. 19. 2008

To:  
Dr. Michael Kidd  
Editor-in-Chief  
Journal of Medical Case Report

Dear Dr. Kidd;  
    Sub: Submission of revised version of manuscript “Valsalva Retinopathy In Pregnancy”

We are submitting the revised version of the above-mentioned manuscript for consideration for publication in the “Journal of Medical Case Report”
We thank the reviewers for their valuable comments. We trust you will find our efforts to revise the manuscript to suit the reviewers’ recommendations adequate. Please find attached the main manuscript and our answers to the Reviewers’ comments.

With regards,
Yours sincerely,

Abdullah Al-Mujaini, MD, FRCSC

Corresponding author

Dr. Abdullah Al-Mujaini  
Department of Ophthalmology  
Sultan Qaboos University Hospital  
123 Al Khod / Muscat  
Sultanate of Oman  
Phone: (968) 24141713  Fax: (968) 24144580  
Email: mujainisqu@hotmail.com
Answers to Reviewers’ Comments

Reviewer 1

Comment 1. This is generally well-written paper which is very clear and succinct in its description of the patient’s presentation and management

Reply: Thank you.

Comment 2. Perhaps discussing the potential mechanisms that predispose to Valsalva retinopathy in pregnancy may help strengthen it for publications.

Reply: Done.

Comment 3. Needs some language corrections before being published.

Reply: The language corrections are made

Reviewer 2

Comment 1. A nice, coherent, and interesting case presentation of Valsalva retinopathy in pregnancy.

Reply: Thank you

Comment 2. The authors change the last sentence of the conclusion where they state that “we can confirm that the risk of recurrence is low”. They can only confirm that the risk of recurrence is not 100%.

Reply: The Conclusion part is rephrased.